Author's response to reviews

Title: BRCA1 mutations in women with familial or early-onset breast cancer and BRCA2 mutations in familial cancer in Estonia.

Authors:

Kristiina Tamboom (kristiinatamboom@hot.ee)
Krista Kaasik (kkaasik@yahoo.com)
Jelena Aršavskaja (Jelena@celecure.com)
Mare Tekkel (mare.tekkel@tai.ee)
Aili Lilleorg (aili.lilleorg@regionaalhaigla.ee)
Peeter Padrik (Peeter.Padrik@kliinikum.ee)
Andres Metspalu (andres@ebc.ee)
Toomas Veidebaum (toomas.veidebaum@tai.ee)

Version: 4 Date: 10 March 2010

Author's response to reviews:

Dear Hereditary Cancer in Clinical Practice Editorial Team,

Please find our article:” BRCA1 mutations in women with familial or early-onset breast cancer and BRCA2 mutations in familial cancer in Estonia".

We have improved English language details as editor suggested.

With kind regards,
Kristiina Tamboom
Corresponding author on behalf of the all co-authors.